Author's response to reviews

Title: The impact of different doses of vitamin A supplementation on male and female mortality. A randomised trial from Guinea-Bissau.

Authors:

Dorthe Yakymenko (dortheyakymenko@gmail.com)
Christine S Benn (cb@ssi.dk)
Cecario Martins (c.martins@bandim.org)
Birgitte R Diness (Birgitte@diness.dk)
Ane Fisker (a.fisker@bandim.org)
Amabelia Rodrigues (a.rodrigues@bandim.org)
Peter Aaby (p.aaby@bandim.org)

Version: 6 Date: 22 August 2011

Author's response to reviews: see over
Dear Editor,

We are very grateful for the positive response after our revision.

Regarding the comment on “Written and verbal consent…” being changed to “Verbal consent…” we found during our revision that our previous description was not correct. Since the majority of the population is by all practical means illiterate, information and consent was only verbal. The mothers were offered a written explanation of the purpose of the study, but since it was only wanted on very few occasions, we have found it unnecessary to include this in the article. We have made changes to the method section accordingly.

We hope that this explanation and our changes in the method section are acceptable.

Best wishes

Dorthe Yakymenko